A Bibliography of Papers in Lecture Notes in Computer Science (2008)

Nelson H. F. Beebe University of Utah Department of Mathematics, 110 LCB 155 S 1400 E RM 233 Salt Lake City, UT 84112-0090 USA Tel: +1 801 581 5254 FAX: +1 801 581 4148

E-mail: beebe@math.utah.edu, beebe@acm.org,

          beebe@computer.org (Internet)
WWW URL: http://www.math.utah.edu/~beebe/

14 October 2017 Version 1.04

Title word cross-reference

2 [6].

754r [7].

8th [18].

According [12]. Algebraic [4]. Algorithms [17]. Analysis [4, 11]. Application [16]. Arithmetic [6].

Banff [17]. Based [3]. between [14]. Bipartite [13].

Canada [17]. Cases [7]. Cincinnati [19]. Collaborative [16]. Coloring [15].

Complexity [4]. Computation [12]. Computational [18]. Conditions [10]. conference [18]. Convex [3]. Cryptography [19]. Curves [9].

December [17]. decimal64 [7]. Definition [11]. degree [14].

EGC [12]. Enclosures [3]. Error [11]. Experimentation [4]. Exponential [7]. Exposing [15].

Format [7]. fourth [17]. Function [7].

Geometric [10]. graph [17]. Graphs [13].

Hierarchical [3]. Homograph [15]. Homotopy [10]. 1

ICCS [18]. IEEE [7]. III [18]. Image [13]. Improvement [1]. in-degree [14]. Intentions [15]. International [19, 17, 18]. Interpolation [11]. Interval [2, 3, 6]. Interval-Based [3]. IVP [1].

June [18].

Kraków [18].

Learning [16]. Library [2].

Mechanical [1]. Methods [5]. Mission [2]. Modeling [1]. Modelling [5, 11]. models [17].

Neighborhoods [9]. November [17]. Numbers [4].

Obfuscation [15]. Object [3]. October [19]. OH [19].

PageRank [13, 14, 16]. papers [17]. Part [18]. Performance [1]. Poland [18]. Polyhedral [3]. Post [19]. Post-Quantum [19]. PQCrypto [19]. Practice [9]. Probabilistic [14]. Proceedings [18, 19].

Quantum [19]. Queries [10].

Randomness [8]. Real [4, 12]. relation [14]. Representations [3]. revised [17]. Robustness [8].

science [18]. Second [19]. SmartMOBILE [1]. Solid [11]. Spline [9]. SSE [6]. SSE-2 [6]. Stochastic [5]. Strings [15]. Subroutine [2]. Systems [1].

Tagging [13]. Technique [16]. Theory [9, 12]. Tolerant [10]. Topological [9]. Traffic [5]. Transfinite [11].

Unicode [15]. USA [19]. Using [13, 6].

ValEncIA [1]. ValEncIA-IVP [1]. Validated [1]. Verified [5].

WAW [17]. web [17]. web-graph [17]. Well [11]. Well-Definition [11]. Workshop [19, 17]. Worst [7].

References

Auer:2008:VMM

[1] Ekaterina Auer, Andreas Rauh, Eberhard P. Hofer, and Wolfram Luther. Validated modeling of mechanical systems with SmartMOBILE: Improvement of performance by ValEncIA- IVP. Lecture Notes in Computer Science, 5045:1–27, 2008. CODEN LNCSD9. ISSN 0302-9743 (print), 1611- 3349 (electronic). URL http://link.

springer.com/content/pdf/10.1007/
978-3-540-85521-7_1.pdf; http://
link.springer.com/content/pdf/bfm:
978-3-540-85521-7/1.pdf.

Corliss:2008:ISL

[2] George F. Corliss, R. Baker Kearfott, Ned Nedialkov, John D. Pryce, and Spencer Smith. Interval subroutine library mission. Lecture Notes in Computer Science, 5045:28–43, 2008. CODEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (elec-

tronic). URL http://link.springer.
com/content/pdf/10.1007/978-3-540-
85521-7_2.pdf.

Dyllong:2008:CPE

[3] Eva Dyllong. Convex polyhedral enclosures of interval-based hierarchical object representations. Lecture Notes in Computer Science, 5045:44–56, 2008. CO- DEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (electronic). URL http:/

/link.springer.com/content/pdf/10.
1007/978-3-540-85521-7_3.pdf.

Emiris:2008:RAN

[4] Ioannis Z. Emiris, Bernard Mourrain, and Elias P. Tsigaridas. Real algebraic numbers: Complexity analysis and experimentation. Lecture Notes in Computer Science, 5045:57–82, 2008. CO- DEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (electronic). URL http:/

/link.springer.com/content/pdf/10.
1007/978-3-540-85521-7_4.pdf.

Kempken:2008:VMS

[5] Sebastian Kempken and Wolfram Luther.

Verified methods in stochastic traffic modelling. Lecture Notes in Computer Science, 5045:83–101, 2008. CODEN LNCSD9. ISSN 0302-9743 (print), 1611- 3349 (electronic). URL http://link.

springer.com/content/pdf/10.1007/
978-3-540-85521-7_5.pdf.

Lambov:2008:IAU

[6] Branimir Lambov. Interval arithmetic using SSE-2. Lecture Notes in Computer Science, 5045:102–113, 2008. CODEN LNCSD9. ISSN 0302-9743 (print), 1611- 3349 (electronic). URL http://link.

springer.com/content/pdf/10.1007/
978-3-540-85521-7_6.pdf; http://
www.springerlink.com/content/576312312g511037/

. Lefevre:2008:WCE

[7] Vincent Lefèvre, Damien Stehlé, and Paul Zimmermann. Worst cases for the exponential function in the IEEE 754r decimal64 format. Lecture Notes in Computer Science, 5045:114–126, 2008. CO- DEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (electronic). URL http:/

/link.springer.com/content/pdf/10.
1007/978-3-540-85521-7_7.pdf.

Michelucci:2008:RR

[8] Dominique Michelucci, Jean Michel Moreau, and Sebti Foufou. Robustness and randomness. Lecture Notes in Computer Science, 5045:127–148, 2008. CO- DEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (electronic). URL http:/

/link.springer.com/content/pdf/10.
1007/978-3-540-85521-7_8.pdf.

Miller:2008:TNS

[9] Lance Edward Miller, Edward L. F.

Moore, Thomas J. Peters, and Alexander Russell. Topological neighborhoods for spline curves: Practice & theory. Lecture Notes in Computer Science, 5045: 149–161, 2008. CODEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (elec-

tronic). URL http://link.springer.
com/content/pdf/10.1007/978-3-540-
85521-7_9.pdf.

Shapiro:2008:HCT

[10] Vadim Shapiro. Homotopy conditions for tolerant geometric queries. Lecture Notes in Computer Science, 5045:162– 180, 2008. CODEN LNCSD9. ISSN 0302- 9743 (print), 1611-3349 (electronic). URL

http://link.springer.com/content/
pdf/10.1007/978-3-540-85521-7_10.
pdf.

Stewart:2008:TIW

[11] Neil F. Stewart and Malika Zidani. Transfinite interpolation for well-definition in error analysis in solid modelling. Lecture Notes in Computer Science, 5045:181– 192, 2008. CODEN LNCSD9. ISSN 0302- 9743 (print), 1611-3349 (electronic). URL http://link.springer.com/content/
pdf/10.1007/978-3-540-85521-7_11.
pdf.

Yap:2008:TRC

[12] Chee Yap. Theory of real computation according to EGC. Lecture Notes in Computer Science, 5045:193–237, 2008. CO- DEN LNCSD9. ISSN 0302-9743 (print), 1611-3349 (electronic). URL http:/

/link.springer.com/content/pdf/10.
1007/978-3-540-85521-7_12.pdf.

Bauckhage:2008:ITU

[13] Christian Bauckhage. Image tagging using PageRank over bipartite graphs. In Anonymous, editor, Proceedings of the 30th DAGM Symposium on Pattern Recognition, volume 5096 of Lecture Notes in Computer Science, pages 426– 435. Springer-Verlag Inc., New York, NY, USA, 2008. ISBN 3-540-69320-3. ISSN 0302-9743 (print), 1611-3349 (electronic). LCCN ????

Litvak:2008:PRB

[14] Nelly Litvak, Werner R. W. Scheinhardt, and Yana Volkovich. Probabilistic relation between in-degree and PageRank. In Aiello et al. [17], pages 72–83. ISBN 3-540-78808-5, 3-540-78807-7. ISSN 0302-9743 (print), 1611-3349 (electronic). LCCN QA76.9.A43 W39 2006.

Wenyin:2008:EHO

[15] L. Wenyin, A. Y. Fu, and X. Deng. Exposing homograph obfuscation intentions by coloring Unicode strings. In Yanchun Zhang, editor, Asia–Pacific web conference: Progress in WWW research and development: APWeb 2008, volume 4976 of Lecture Notes in Computer Science, pages 275–286. Springer-Verlag Inc., New York,

NY, USA, 2008. ISBN 3-540-78848-4. ISSN 0302-9743 (print), 1611-3349 (electronic). LCCN ????

Yang:2008:APT

[16] Shenggang Yang, Jianmin Zhao, Xueyan Zhang, and Limei Zhao. Application of PageRank technique in collaborative learning. In Anonymous, editor, Advances in Blended Learning: Second Workshop on Blended Learning, WBL 2008, Jinhua, China, August 20–22, 2008. Revised Selected Papers, volume ???? of Lecture Notes in Computer Science, pages 102– 109. Springer-Verlag Inc., New York, NY, USA, 2008. ISSN 0302-9743 (print), 1611- 3349 (electronic).

Aiello:2008:AMW

[17] William Anthony Aiello et al., editors. Algorithms and models for the webgraph: fourth international workshop, WAW 2006, Banff, Canada, November 30–December 1, 2006: revised papers, volume 4936 of Lecture Notes in Computer Science. Springer-Verlag Inc., New York, NY, USA, 2008. ISBN 3-540- 78808-5, 3-540-78807-7. ISSN 0302-9743 (print), 1611-3349 (electronic). LCCN QA76.9.A43 W39 2006.

Bubak:2008:CSI

[18] Marian Bubak, Geert Dick van Albada, Jack Dongarra, and Peter M. A. Sloot, editors. Computational science – ICCS 2008. 8th international conference, Kraków, Poland, June 23–25, 2008. Proceedings, Part III, volume 5103 of Lecture Notes in Computer Science. Springer- Verlag Inc., New York, NY, USA, 2008.

Buchmann:2008:PQC

[19] Johannes Buchmann and Jintai Ding, editors. Post-Quantum Cryptography: Second International Workshop, PQCrypto 2008 Cincinnati, OH, USA, October 17–19, 2008 Proceedings, volume 5299 of Lecture Notes in Computer Science. Springer-Verlag Inc., New York, NY, USA, 2008. ISBN 3-540-88403-3. LCCN ????